

Financial impact of regulatory sanctions on listed companies

Laure de Batz^{1,2}

Published online: 27 February 2020 © Springer Science+Business Media, LLC, part of Springer Nature 2020

Abstract

We examine the impact of the enforcement of financial regulations by the French Financial Market Authority on sanctioned firms. The early stages of the enforcement process are by law confidential, with an internal investigation and bilateral exchanges between the defendant and its regulator. The public hearing by the Enforcement Committee leads to a single publication of the decision, being the only public communication. Using an event study methodology, we find that the confidentiality of the initial steps of enforcement procedures is respected and that markets account for the publication of sanctions. Still, reactions are limited in absolute and relative terms, both compared to past studies and in terms of reputational penalty. Some parameters trigger a stronger reaction, but not the most straightforward (such as the cash fine or behavioral sanction). The results echo the reputation for leniency of sanctions (scarce procedures, lax verdicts, low fines, ending neglected by analysts and investors), despite consecutive regulatory tightenings and long procedures. They question the efficiency of enforcement.

Keywords Sanction \cdot Fraud \cdot Event study \cdot Regulation \cdot Information and market efficiency \cdot Listed companies

JEL Classification $~G14 \cdot G18 \cdot K42 \cdot N24$

[☑] Laure de Batz 87537751@fsv.cuni.cz; laure.debatz@gmail.com

¹ Faculty of Social Sciences, Institute of Economic Studies, Charles University of Prague (IES FSV UK), Opletalova 26, 110 00 Prague 1, Czech Republic

² Laboratory of Excellence for Financial Regulation (LabEx-ReFi), Maison Des Sciences Économiques, Centre D'Économie de la Sorbonne (CES), University Paris 1 Panthéon-Sorbonne, 106-112 bd de l'Hôpital, 75642 Paris Cedex 13, France

1 Introduction

Regulating financial markets targets diverse objectives: encourage sound and transparent financial markets; deter excessive risk-taking; foster market participants to act responsibly; and compensate for past wrongdoings; etc. Being budgetary constrained (Carvajal and Elliott 2007), regulators focus on the most severe uncovered regulatory breaches. Hence, sanctions are expected to be interpreted as a significant negative information regarding the firm and/or the individual being sanctioned, justifying a reputational cost imposed by market participants. If a potential sanction stands for a credible threat, its mere existence could complement financial regulation by incentivizing market players to abide by the law. Markets would be a channel complementary to enforcement to deter future misconducts and to induce firms, top managers, and individuals to act responsibly (Engelen 2011). All in all, what are the consequences of regulatory enforcement on sanctioned listed firms?

Due to data constraints, limited research was done to date on jurisdictions other than the United States (US) on this question. Regarding France in particular, previous studies focused on one type of regulatory breach (accounting frauds, Djama 2008; or insider trading, Fonteny 2017), or covered few sanctions (47 sanctions of listed companies, Kirat and Rezaee 2019).¹ The objective of this paper is to provide a comprehensive investigation of market reactions to sanctions of listed companies for the French market, by constructing and exploiting a unique dataset of all the sanctions and settlements made by the French Financial Market Authority (AMF). Our sample is improved compared to prior research as it covers exhaustively the sanctions of the AMF since its creation in 2003, until late 2016, based on public and confidential data.

Beyond its novelty, the use of a database pertaining to the French market is particularly relevant since the French enforcement process is highly compatible with an event study. Indeed, the dates of the consecutive steps of the sanction procedures are unique and available (publicly or confidentially). The initial steps of enforcement procedures are confidential, and should not lead to abnormal market reactions.² Conversely, the last two steps (Enforcement Committee hearing and publication of the decision) are public information, with the nature and the size of the penalty being precisely identified on the day of the publication. Hence, they are expected to influence market expectations, should the sanctions signal a negative assessment by the Regulator. Additionally, sanctions are not private information for the firms: they are revealed by the Regulator and exogenous to the firm's agenda. Therefore, there is no

¹ This gap in the literature can be accounted for by the (increasingly) limited open access to data. Indeed, sanction reports are frequently published anonymized (part of the sanction decision of the Enforcement Committee (AMF EC), *i.e.* ex ante) or anonymized ex post at the EC AMF Chairman's discretion. The compounded anonymization rate is 57% (de Batz 2017a, b). Additionally, some dates can be missing in sanction reports. Lastly, in 2018, a regulatory change led to an anonymization of most of the sanctions from 2004 to 2013.

² Like in the United Kingdom (UK, Armour et al. 2017). Conversely, in the US, the trigger event can be early communication by the Regulator and/or the defendant.

self-selection nor optimization process made by the sanctioned companies: the AMF decides independently when to publish its decision.

Following the rich literature on the repercussions of corporate misconducts, an event study methodology investigates for abnormal returns following the milestones of sanction procedures (the "events"). Complementarily, the market value losses are estimated, questioning reputational penalties following the sanction. The results are complemented by cross-sectional multivariate analysis of the determinants of the abnormal returns. Investors should amend their investment strategies proportionally to the severity of the financial misconduct (Choi and Kahan 2007).

This article enriches the understanding of market reactions to enforcement by investigating how regulatory decisions are perceived by market players to the largest possible extent: depending on the procedure (sanction or settlement), on the verdict (sanction, acquittal, anonymization), on the offender's characteristics, on the timing of the enforcement, on the media coverage, and on the legal environment. It questions, over a long and up-to-date time span, potential abnormal returns following the milestones of sanction procedures, from the investigation until the publication of the decision (see Fig. 1). Three reasons make it particularly interesting: (1) only the most serious regulatory breaches detected by the AMF end with a sanction procedure; (2) sanction procedures are long (close to 3 years on average); and (3) firms listed in a stock exchange are likely to be subject to a closer scrutiny by the Regulators and to receive more media attention.

The results indicate, on average, statistically significant abnormal reactions to sanctions. Guilty listed firms incur abnormal losses in returns after the sanction decision and its publication. Still, reactions are limited in absolute and relative terms and no reputational penalty is found. Conversely, no abnormal reaction follows either the ignition of the procedure, or the statement of objection, stressing a compliance with the confidentiality of enforcement procedures. Cross-sectional regressions show that abnormal returns are unrelated to the main features of the sanctions (cash fines, as in Armour et al. 2017, and disciplinary sanctions). Other aspects contribute to negative abnormal returns after the publication: procedures initiated with an investigation, longer procedures, the involvement of the top management (chairman, CEO, CFO, owner of the firm), higher media coverage, being a financial or technological firm, and better economic times. Complementary event studies conclude with no abnormal reaction following anonymized sanctions or settlements, in line with their confidential or less severe natures. Acquittal decisions do not trigger straightforward reactions.

This article hence contributes to the existing literature on the impact of enforcement by detailing the timing and transmission schemes of a sanction procedure into the French stock markets. The results contribute to improving the understanding of financial regulation, and of the reasons why French enforcement and sanctions can be said to be scarce and lenient.³ The current framework can be questioned

³ Sanctions by the AMF suffer from a reputation of being scarce (*i.e.* low probability of being caught) and lenient (*i.e.* lax verdicts with low fines). They do not receive a straightforward coverage by the media nor by financial analysts: most identified misconducts are dealt with bilaterally and confidentially between the AMF and the regulated entity. Even the highest sanction in history (35 million euros sentenced in July 2017) did not cast an unequivocal analysis. In fact, financial penalties are low in abso-



Fig. 1 Timeline of an AMF enforcement action. Sources: AMF, Author

as: (1) the market reactions are limited in absolute and relative terms, (2) the most straightforward features of the sanctions (cash fines and disciplinary sanctions) do not matter, and (3) the fines remain extremely limited compared to maximum legal thresholds (despite consecutive increases along the period under review), though trending upwards (de Batz 2017a, b). The following policy recommendations can be made, under the assumption that a credible and efficient enforcement should be priced in by the markets: (1) more communication from the Regulator along the enforcement process, as done by the US SEC, to help market participants better and more rapidly assimilate the information on the misconducts being investigated and as a tool to educate and set example ("name and shame", as enforced in the UK); (2) more severe and less frequent sanctions (significantly higher fines, closer to the legal maximum, and more disciplinary sanctions), if the Regulator believes that the credibility of a sanction should be measured in the market reactions, as in the US for example;⁴ and (3) more sanctions of individuals (top managers in particular), in order to reinforce accountability and encourage best practices. Better enforcing financial regulations is all the more relevant that market participants are increasingly regulated, partly as a consequence of the Great Financial Crisis. In the end, regulation should support and accompany a healthy development of firms, and not suffocate them. It is a crucial parameter of the attractiveness and the strength of securities markets in terms of fund raising (La Porta et al. 2006), of market capitalization (Beny 2008), and of liquidity (Cumming et al. 2011).

The rest of the article is structured as follows. Section 2 presents a literature review. Section 3 outlines the institutional framework of enforcement in France and formulates the hypotheses. The subsequent Section 4 describes the methodologies of the event study and of the cross-sectional regression, and the data. Section 5 presents the results and Section 6 concludes.

Footnote 3 (continued)

lute and relative terms, despite four-consecutive reinforcements of AMF's enforcement powers since its creation in 2003. For example, the maximum legal fines were repeatedly increased (up to 100 million euros, or 10 times the gains realized for firms, see Table 1). In the end, the translation of sanctions into returns of listed companies is a priori unclear. This marks a sharp difference with other jurisdictions (Anglo-Saxon countries in particular) or with sanctions by other French Regulatory Authorities (such as the Competition Authority).

⁴ For example, in Karpoff et al. (2008a), only 8% of the 585 firms received a fine from their regulatory agencies over the period 1978–2002. The mean was 107 million dollars (60 million when excluding an exceptional case).

2 Literature review

Securities regulation for capital markets and the subsequent enforcement aim at informing investors and at deterring and uncovering white collar crimes. Several tools are at the Regulator's disposal: market surveillance, bilateral exchanges with regulated entities, settlements, and (monetary and/or non-monetary) sanctions, on which this article focusses. Alternative enforcement tools can also encourage best practices (Berger and Davies 1998; Barth et al. 2004; La Porta et al. 2006), such as private enforcement and disclosure of information, in particular in a context of imperfect information (Garoupa 1999).

Rational agents will break the law if the profits derived from their crimes exceed the expected costs. In a seminal contribution, Becker (1968) proposes a theoretical framework for the economics of crime to reach an optimal enforcement (deterrence of future crimes, compensation, and vengeance). In his model, sanctions will circumvent frauds and foster compliance with regulation depending on three parameters: (1) the expected profits from committing the fraud (*i.e.* the harm inflicted upon victims or the society, justifying a sanction); (2) the probability of being caught;⁵ and (3) the subsequent the total cost for being caught (i.e. the cumulated costs of punishment including fines, disciplinary sanctions, jail, higher financing costs, reputational penalty, etc.). This article focusses on this third parameter. Indeed, for firms, sanction procedures are a major legal risk as they convey direct and indirect financial consequences: long and costly procedures, the cash fine set by the Regulator, second-round effects such as higher costs of funding and doing business,⁶ and possibly a "reputational penalty" from the market. Hence, the share price is expected to contract after a sanction, though some contrarian forces may play. Some investors may fail to or decide not to react to the news, while risk-seeking investors could search for investments in firms more prone to play with the limits of the law, possibly synonym of higher returns. Most past literature concludes that the reputational penalty must be accounted both when setting policy standards and when making business decisions (Karpoff et al. 2008b). On the one hand, financial markets could complement enforcement as a channel to induce firms and market participants to

⁵ The (actual or perceived, Garoupa 1999) detection rate by the Regulator (or by other market participants) is low, even though misconducts on financial markets are frequent. The probability of being caught depends on the public expenditures on enforcement, courts, police, etc. They are by nature constrained. No data exists on frauds which went undetected. Bussmann and Werle (2006) estimated, in the global survey, that only 4% of the detected economic crimes were identified by law Enforcement Agencies, most of them being detected by the firms themselves. On average, only 2 to 5% of the American listed companies are investigated per year by the Securities and Exchange Commission (US SEC), according to Cumming and Johan (2013). Drake et al. (2014) stressed that, from 1996 to 2004, out of the 15% of large American publicly traded firms engaged in fraud each year, only 4% are in the end detected.

Similarly, the detection rates for cartels are low, despite being larger in terms of scope and duration. In the US, Bryant and Eckard (1991) estimated the annual probability of being caught for a cartel from 13 to 17%. In Europe, Combe et al. (2008) estimated it from 12.9 to 13.1% from 1969 to 2008, based on European Commission data.

⁶ The sanctioned entity can face higher insurance premia, more expensive funding conditions, tougher client relationships, and additional investments to compensate for the demonstrated market failure (IT, process improvements, marketing, and communication, etc.).

behave responsibly (Engelen 2011). On the other hand, the threat of a reputational penalty from the market, exceeding by far the legal sanction, could deter regulatory breaches (for the US, Karpoff and Lott 1993; Karpoff et al. 2008a; for the UK, Armour et al. 2017). Otherwise, the perceived under-punishment of fraud might encourage financial misconduct. The question is then whether financial misconduct pays, if expected profits from regulatory breach(es) exceed the total cost of a sanction (monetary (fines) and non-monetary (reputation) costs), for a given probability of being caught.

The impact of regulatory sanctions on the behavior of financial investors was empirically studied by the literature for numerous jurisdictions,⁷ and from different angles, either for given populations,⁸ for specific regulatory breach(es),⁹ or depending on the media coverage.¹⁰ They echo a long literature on corporate regulatory breaches.¹¹ Some investigated the difference between first-time and repeated offenses, with higher market corrections (Gondhalekar et al. 2012). The most studied country is the US,¹² thanks to higher transparency from Regulators and defendants (along the enforcement procedures), the size of the market, and the easy availability of a wide range of data on financial fraud. The consecutive steps of the enforcement procedures were studied, typically with event studies.¹³ Assuming financial markets are informationally efficient (Fama et al. 1969), all the available information (here the regulatory sanctions) should be reflected immediately by the market (in the stock prices of the sanctioned listed companies). Past researches show that US markets react significantly to sanctions, in particular to the earlier stages of the procedures (Feroz et al. 1991; Pritchard and Ferris 2001). Still, in an in-depth comparative study, Karpoff et al. (2014) stressed that the consecutive nature of the US enforcement process significantly biases abnormal returns estimates. Similar event studies were conducted following the news of a financial frauds and regulatory sanctions for

⁷ Such as Canada, China, France, Germany, Japan, Luxembourg, Malaysia, the Netherlands, and the UK.

⁸ Such as listed companies (Karpoff and Lott 1993; Kirat and Rezaee 2019), banks (Köster and Pelster 2017; Caiazza et al. 2018), asset managers (Choi and Kahan 2007), or the management of the firm of the sanctioned firm (Karpoff et al. 2008b).

⁹ Such as financial and accounting frauds (for France, Djama 2008), the accounting disclosure (Karpoff et al. 2008b), or insider trading news (Rogers et al. 2016; Fonteny 2017).

¹⁰ This article focusses on the role of dissemination of information of the press, and not on the creation (Drake et al. 2014). The coverage by mass media can alleviate information problems for listed firms (Fang and Peress 2009), in particular for individual investors (Fang et al. 2014). Miller (2006) demonstrated that an accounting fraud is more likely to be echoed in the press for firms which already receive more attention from the press. (Past researches typically conclude with higher market reactions. Rogers et al. (2016) showed that the media plays an economically important role in price formation in securities markets, by widening the dissemination of insider trading disclosures.

¹¹ A wide range of regulatory breaches can damage one's reputation: financial fraud, corporate malfeasance (anti-trust violation, bribery, tax evasion, illegal political contributions, employer discrimination, etc.), false or misleading advertising, product recalls, airplane accidents, environmental violations, illicit allegations, etc. Their impacts are typically investigated with an event study methodology.

¹² Ordered chronologically: Feroz et al. 1991; Karpoff and Lott 1993; Alexander 1999; Pritchard and Ferris 2001; Karpoff et al. 2008a; Tibbs et al. 2011; Haslem et al. 2017.

¹³ Accounting and Auditing Enforcement (AAER), US SEC formal or informal investigations, Wells Notice issuance, sanctions, and class action filing.

other jurisdictions. They are scarcer, possibly due the data availability challenges. Whatever the country or region under review, these event studies conclude with negative, rapid,¹⁴ and significant abnormal market reactions to such financial news from the Regulator. Still, the extent of the abnormal returns varies substantially, as well as the timing. There can also be some anticipation from the market, possibly resulting from rumors or private information regarding the sanction.¹⁵

Beyond the impact of sanctions on returns (put it differently abnormal returns estimated using an event study methodology), some studies isolate the reputational penalty imposed by the market (if any) from the cost of the sanction.¹⁶ To estimate this reputational penalty, a "residual approach" is typically used. The financial sanction (*i.e.* the fine, and possibly other related costs like financing costs, compensations, etc.) is deducted from the overall estimated abnormal market reaction following the news of the sanction (Jarrell and Peltzman 1985; Karpoff and Lott 1993; Karpoff et al. 2008a; Murphy et al. 2009; Armour et al. 2017). They conclude that the reputational penalty exceeds, by far, the financial sanction set by the Regulator.

Some articles distinguish misconducts depending on the relationship between the offender and the offended, to challenge whether it influences market reactions. The sanctions are split depending on whether the regulatory breach impacted related parties to the offender (investors, employees, customers, suppliers) or third parties (market participants, the public, etc.). They conclude that the reputational cost of wrongdoings against related parties is significantly higher (for the US, Alexander 1999; Karpoff et al. 2008a; Murphy et al. 2009; Tibbs et al. 2011; for the UK, Armour et al. 2017).

Finally, part of the literature discriminates the reactions depending on the content of the decision or the communication by the Regulator: being acquitted or being investigated for alleged financial regulatory breaches. This is particularly relevant to studies on the US, where Regulators and defendants are allowed to (and do) communicate along the enforcement process. Some studies found negative impacts on returns of allegations of financial misconduct (*i.e.* for being investigated by one's authority), demonstrating a reputational penalty to the mere suspicion of misconduct (Feroz et al. 1991; Pritchard and Ferris 2001; Murphy et al. 2009; Nelson et al.

¹⁴ Lin and Rozeff (1995), for example, demonstrated that 85 to 88% of private information is incorporated into prices within one trading day.

¹⁵ For shorter term reactions, out of an extensive literature review of 55 similar studies (*details are available on demand*), the following averages for Cumulative Average Abnormal Returns were estimated:

[•] For the most frequently used event window (55% of the sampled studies, most frequently US), $CAAR_{[-1;+1]} = -7.7\%$, ranging from -25 up to -0.5%, with a standard error of 7.6%;

[•] For shorter event windows: $CAAR_{[0;+1]} = -7.6\%$, ranging from -20.6 up to -0.7%, with a standard error of 7.6%; $CAAR_{[-1;0]} = -3.2\%$, ranging from -6.3 up to -0.8%, with a standard error of 2.3%;

[•] For longer short term event windows ([-2;+2], [-3;+3], [-5;+5], and [-7;+7]), $CAAR_{[-i;+i]} = -9.4\%$, ranging from -19 up to -1%, with a standard error of 7.6%.

These averages are higher with the two past studies on France $(CAAR_{[-1;+1]} = -1.05\%)$ and $CAAR_{[-1;+1]} = -2.98\%$, not statistically significant), as well as short-term estimates on Europe, the UK, and Germany (-1.55%) in CAAR.

¹⁶ Reputation is a key asset and deserve investments (Fiordelisi et al. 2014; Heidinger and Gatzert 2018).

2009; Dyck et al. 2010; Tibbs et al. 2011; Haslem et al. 2017). The first announcement of an alleged (potentially sanctioned) regulatory breach triggers the highest and most significant negative market reaction.

3 The French institutional framework of enforcement and the research questions

3.1 Sanctioning powers of the AMF

As part of its mandate, the Enforcement Committee of the French Financial Market Authority (AMF EC) sanctions market players which infringe the sets of rules they are subjected to: the Monetary and Financial Code, and the AMF General Regulation. The goal of sanctions, from a regulatory point of view, is to strengthen the marketplace, by condemning wrongdoings and setting examples. Four main regulatory breaches are sanctioned by the AMF (see Table 1): three market abuses: (1) breaches of insider dealing regulations (the use and/or divulgence of insider information for investment decisions); (2) price manipulations (a deliberate misconduct to influence securities prices and fair price formation); (3) breaches of public disclosure requirements (a failure to comply with financial reporting laws and regulations); and (4) any breach of the Monetary and Financial Code and the AMF General Regulation (a failure to meet with professional obligations). From 2004, when the AMF first sanctioned after its creation in 2003, to 2016, 308 decisions were made and published on the AMF website. They stood for 196 million euros of cumulated fines.¹⁷ Until late 2016, for a given regulatory breach(es), such administrative procedures could be conducted by the AMF, in parallel to criminal prosecutions. All procedures follow the same four milestones (see Fig. 1). A sanction decision can be comprised of cash fines,¹⁸ disciplinary sanctions,¹⁹ and its publication.²⁰ The offender (firm and/or individual) and/or the AMF Chairman of the Board can appeal the decision towards

¹⁷ 24 sanctions were made per year on average, to which add 6 settlements per year since 2012, when this new procedure was first concluded. When excluding the 9% acquittals, 94% of the guilty sanctions included a cash fine, for an average 688,320 euros, paid to the French Treasury (or to the guarantee fund to which the professional belongs).

¹⁸ There is neither binding rule nor clear guidelines on how to value fines. Time consistency and the maximums set legally are the two key objective parameters to set a fine, to which add specificities of the respondent (gravity and duration of the financial misconduct(s), financial situation of the defendant, magnitude of the obtained gains or advantages, losses by third parties, etc.). Maximum fines were increased three times over the period under review and can amount up to 100 million euros for market abuses committed by professionals, or 10 times any profit.

¹⁹ (1) Warning and/or blame, depending on the seriousness of the wrongdoing(s); and (2) "ban on activity", covering temporary or permanent ban on providing some or all services, suspension or withdrawal of professional license, and temporary or permanent ban on conducting some or all businesses.

²⁰ Most sanctions are published, in particular in recent years, except if such disclosure would seriously jeopardize the financial markets or cause disproportionate damage to the parties involved. The EC decides whether to publish its decision, where to publish it (mostly on the French Official Journal for Legal Notices (BALO) and on the website of the AMF, possibly in a given set of magazines, at their expenses) and whether to anonymize it (entirely or partially).

Table 1 Taxonomy of the sanctions

Which Procedure for W	'hat Regulatory Breach?
Investigation	1. Market abuses:
	1.1. Insider dealing (29% of the sample ¹): use or dissemination of information which is not publicly available to other investors for personal gain (for example, a director with knowledge of a takeover bid)
	1.2. Price manipulation (10% of the sample ¹): distortion of the price-setting mechanism
	1.3. Dissemination of false or misleading information (about a company's financial circumstances) <i>vis-à-vis</i> the Regulator or investors (63% of the sample ¹)
Control	2. Any failure to meet the professional obligations set out in the Monetary and Financial Code, the AMF General Regulation and AMF-approved profes- sional rules (38% of the sample ¹)
Who?	
	1. Any professionals under AMF supervision if they breach their professional obligations under the law, regulations and professional rules approved by the AMF
	2. Individuals acting under the authority or on behalf of these professionals
	3. Any person that commits market abuse, or any other breach that could impair investor protection or interfere with orderly markets
By How Much?	
	Up to 100 million euros (or 10 times any profit earned) for professionals under the AMF supervision
	Up to 300,000 euros (or 5 times any profit earned) for individuals acting under the authority (or on behalf of a professional) for failing to meet professional obligations
	Up to 15 million euros (or 10 times any profit earned) for individuals acting under the authority (or on behalf of a professional) for market abuses
	Up to 100 million euros (or 10 times any profit earned) for other persons (issuers and their executives, auditors, others) for market abuses

This table lists the regulatory breaches which can be sanctioned by the AMF, through which proceeding, who can be sanctioned with what financial risk. Listed companies, and their employees, can be investigated or controlled for any market abuse or failure to meet their professional obligations. They risk maximum cash fines of 100 million euros, possibly assorted with behavioral sanctions (warning, blame, or bans on activity). The maximum fine ever set was 35 million euros, in 2017

Sources: AMF, Author - See de Batz (2017a, b) for details

¹Several breaches can be sanctioned per sanction

four different jurisdictions: the Court of Appeal of Paris, the Court of Cassation, via priority preliminary ruling on constitutionality, and the State Council.

An originality of this article is that it exploits a crucial feature of the French financial regulatory framework. All the enforcement process is, by law, confidential until the Enforcement Committee hearing and the subsequent publication of the sanction decision. Consequently, like in the UK until recently (Armour et al. 2017), there are no private litigation nor class action claims occurring over the enforcement process. Additionally, and contrary to the US, newspaper articles or a whistleblowing never triggered an enforcement procedure sanctioned by the AMF over the period under review. All (but one in 2017) sanctioned infringements were identified by the AMF.²¹ Hence, a priori, no speculation on stock returns should be made regarding future financial penalties. Still, this article is an opportunity to investigate for potential leakages from part of the AMF (when the procedure starts) and of the defendants (after the statement of objection).

Within this framework, the AMF's legal attributes to sanction have significantly evolved since 2004. On four occasions, its sanction powers were reformed, broadened and reinforced (de Batz 2017a, b). Additionally, an alternative procedure to sanctions, the settlement proceeding, was introduced in 2010, and first applied in 2012. The latter implies simpler and shorter procedures, initially only for the less serious regulatory breaches (failures to meet with professional obligations), without guilty plea from the offender or appeal possibility. Settlements reduce the costs and risks inherent to a trial, but dilute deterrence from an enforcement perspective. They are clearly preferred by firms (Bussmann and Werle 2006), when the Regulator offers the option. The two latest reforms were enforced in 2016 and have impacted enforcement since 2017.²² They reorganized legal proceedings and reinforced the sanction powers of the AMF. Therefore, such evolutions make it particularly interesting to assess the impact of sanctions on investors from the first sanction pronounced in 2004 until late 2016, before a new set of tougher rules starts to apply.

3.2 Testable hypotheses

This article contributes to answering to a global research question, based on the chronology of sanction procedures: what are the consequences of regulatory enforcement on sanctioned listed companies? To do so, we investigate the informational content of sanction procedures by testing the following hypotheses:

²¹ In the USA a significant share of financial scandals are revealed by the press (Choi and Kahan 2007), associated with a statistically significant impact on prices (Miller 2006).

Conversely, in France, the press is mostly a re-broadcaster of scandal news detected by the regulator (and not a producer of news), hence improving the dissemination of information among actual stakeholders and potential investors and contributing to the efficiency of stock markets (Fang and Peress 2009; Fang et al. 2014).

 $^{^{22}}$ Law on market abuses of 21, June 2016 (Law no. 2016-819) and Law on transparency, the fight against corruption and modernized business life, of 9, December 2016 (Law no. 2016-1691, IV Art. 42-46)

The main changes include: (1) The maximum fine remains 100 million euros but can stand for up to 15% of the annual turnover for a legal entity and has been increased up to 15 million euros or ten times any profit earned for an individual failing to meet his professional obligations. (2) The ban from activity can now exceed 10 years. (3) The powers of the Enforcement Committee have also been broadened to public offerings of unlisted financial instruments (without prospectus) and to crowdfunding. (4) The scope of regulatory breaches eligible to settlement procedures has been widened to all market abuses (insider dealing, price manipulation and dissemination of false information), and no longer only the failures of regulated professions to meet professional obligations. (5) Finally, any decision published on the AMF website should remain online at least for 5 years (which was already the case), but any reference to personal data should be anonymized after 5 years (which was only partially the case).

Hypothesis H_1 The confidentiality of the first two steps of the procedures is respected. No abnormal returns follow the ignition of an investigation or a control (*i.e.* from part of the AMF) or the statement of objections (*i.e.* from part of the AMF) or of the company/individual being investigated or controlled).

Hypothesis H_2 Financial markets matter about the negative signal sent by the Regulator when sanctioning financial misconduct. The sanction and its publication lead to negative abnormal returns for sanctioned firms, due to the fine imposed, the downward revision of forecasts, and possibly a reputational penalty.

Hypothesis H_3 Sanctioned firms undergo a reputational loss for being sentenced guilty. The publication of the sanction decision is associated with negative abnormal returns, exceeding the fine imposed by the Regulator.

Hypothesis H_4 The informational content of the regulatory decision and/or the characteristics of the sanctioned firms impact market reactions.

4 Methodology and data

4.1 Methodology

A standard event-study methodology (MacKinlay 1997; Kothari and Warner 2008) is used to investigate the information content of the four main steps of the AMF enforcement procedure (*i.e.* the "events"). The events are assumed exogenous to the firms: enforcement procedures are independent regulatory decisions, and unrelated to corporate agendas.²³ The impact of the event is measured as the daily Abnormal Returns (AR) of the company being sanctioned around the event, by comparing "actual" ex post returns with "normal" estimated returns. The abnormal returns are taken as unbiased estimates of the total financial consequences of the sanction (all expected uninsured future costs, including reputational losses). Under the null hypothesis H_0 , the "event" has no impact on the distribution of returns for the sanctioned firms (mean or variance effect). A market model, augmented with a sectoral index,²⁴ describes the behavior of returns. The model assumes a jointly multivariate normal and temporally independent distribution of returns. On every day *t* of the event window [-10; +120] including the event day (t = 0), the deviation in an individual stock's daily return from what is expected is taken as an unbiased estimate of

²³ Contrary to events such as annual and quarterly publication, or profit warnings. The exogeneity is also supported by the fact that some sanctions were, in the end, excluded from the sample due to confounding events such as the publication of the results from another judicial procedure or M&As.

²⁴ The results of the event studies are robust when using a market model not adjusted for the sectors, though lower. Detailed results are available in de Batz (2018). Controlling for the sector is supported by the long period under review (2004–2016, including the Great Financial Crisis), and the wide range of sectors of the sanctioned firms.

the financial impact of the "event" on the stock *i* in *t*. This abnormal return $AR_{i,t}$ is defined as:

$$AR_{i,t} = R_{i,t} - \hat{\alpha}_i - \hat{\beta}_i R_{m,t} - \hat{\gamma}_i R_{s_i,t}$$
(1)

where $R_{i,t}$, $R_{m,t}$ and $R_{s_i,t}$ are the returns on day *t* respectively on the stock *i*, on the market portfolio, and on the sector portfolio s_i of company $i.^{25,26} \hat{\alpha}_i$, $\hat{\beta}_i$ and $\hat{\gamma}_i$ are the Ordinary Least Squares (OLS) estimates for every sanction *i* over the estimation window [-120; -11]. To draw overall inferences for the event of interest, abnormal returns are cumulated over time $[t_1;t_2]$ and averaged across sanctions to get the Cumulative Average Abnormal Returns (*CAAR* $[t_1;t_2]$), including the event (see specification (2)). All the sanctions are treated as a group.

$$CAAR_{[t_1;t_2]} = \frac{1}{n} \sum_{i=1}^{n} CAR_{i,[t_1;t_2]} = \frac{1}{n} \sum_{i=1}^{n} \sum_{t=t_1}^{t_2} AR_{i,t}.$$
 (2)

Complementarily, for every sanctioned firm *i*, the shareholders' loss (or gain) $SL_{i;[t_1;t_2]}$ is estimated over $[t_1;t_2]$ by multiplying the market capitalization of the firm *i* on the day preceding the event window $(t_1 - 1) MV_{i,(t_1-1)}$ (in euros) by the $CAR_{i,[t_1;t_2]}$. Shareholder losses are then averaged across the *n* sanctions $(ASL_{[t_1;t_2]})$:

$$ASL_{[t_1;t_2]} = \frac{1}{n} \sum_{i=1}^{n} SL_{i,[t_1;t_2]} = \frac{1}{n} \sum_{i=1}^{n} CAR_{i,[t_1;t_2]} x MV_{i,(t_1-1)}$$
(3)

The (net) average reputational loss (or gain) $RL_{i,[t_1;t_2]}$ for firm *i* is proxied with a residual approach (Jarrell and Peltzman 1985; Karpoff and Lott 1993; Karpoff et al. 2008a; Murphy et al. 2009; Armour et al. 2017). Sanctions are published long after the misconducts and their capitalization in prices, as enforcement procedures last for years. Hence, this wealth loss is not added to the regulatory fines (Armour et al. 2017).²⁷ The financial penalty FP_i for sanction *i* only equals the fine imposed by the Regulator. It is deducted from the abnormal shareholder loss (or gain) due to the event $SL_{i;[t_1;t_2]}$:

$$RL_{i,[t_1;t_2]} = SL_{i,[t_1;t_2]} - FP_i = CAR_{i,[t_1;t_2]} xMV_{i,t_1-1} - FP_i$$
(4)

Finally, cross-sectional tests investigate the link between the magnitude of the abnormal returns after the event (*i.e.* loss or gain incurred by shareholders) and the

²⁵ Equity returns are defined as the daily log difference in value of the equity (including reinvested dividends).

 $^{^{26}}$ Given the wide range of size of sanctioned companies, the broadest benchmark index for the French stock markets (CAC All Shares) proxies the market portfolio. Euronext main sectors are used for each firm. The 10 main sectors are: financials (38% of the sample); industrials (15%); technology (13%); consumer goods (8%); consumer services (8%); health care (6%); basic materials (6%); telecoms (2%); utilities (2%); oil and gas (2%).

²⁷ Conversely, in Karpoff et al. (2008a), the reputational penalty equals the expected loss in present value of future cash flows, due to lower sales and higher contracting and financing costs.

features of the events (see Table 2). It is particularly interesting given the multiple possible causes for abnormal returns: do higher fines, disciplinary sanctions, appeals, recidivism, higher media coverage, more liquid stocks, etc. lead to more negative abnormal returns? A cross-sectional regression for $CAR_{i;[t_1;t_2]}$ on sanctions characteristics is estimated using the usual OLS, with White-corrected standard errors:

$$CAR_{i;[t_1;t_2]} = \delta_0 + \delta_1 x_{i,1} + \delta_2 x_{i,2} + \dots + \delta_m x_{i,m} + \mu_i, \text{ where}E(\mu_i) = 0$$
(5)

where $x_{i,j}$, for j = 1, ..., m, are the *m* characteristics of the *i*th observation, δ_j , for j = 0, ..., *m*, are the m + 1 parameters of the model, and μ_i is the zero-mean disturbance term, uncorrelated with the explanatory variables $x_{i,j}$. Heteroskedasticity-consistent t-statistics will be derived using White-corrected standard errors (MacKinlay 1997).

4.2 Description of the data

A unique dataset was constructed covering the 308 publicly available sanction decisions published on the AMF website over the period 2004–2016. It was completed with a second dataset comprising the 32 settlement decisions made from 2012 to 2016. Most variables were extracted or created from the online sanction reports. They were supplemented by publicly available information and by confidential information, shared by the AMF. The latter covers the names of the entities, for anonymized sanction reports (either ex ante or ex post),²⁸ some missing dates of procedure, and information on sanctions dating back to before the AMF creation (such as sanctions by AMF's forefathers). Finally, softwares were used for market data (Thomson Reuters), and for media coverage (Factiva).

The dataset includes more than 40 variables (see descriptive statistics in Table 2): (1) the characteristics of the sanction (or settlement) procedure (including the type of procedure at the origin with an investigation or a control, the sanctioned regulatory breaches,²⁹ the dates of four milestones of the procedure³⁰); (2) the main features of the verdict (acquittal, cash fine(s), disciplinary sanction(s), ban(s) on activity, anonymization of the sanction,³¹ the chairman of the AMF EC, the length of the sanction report, appeal characteristics,³² whether other listed companies were

²⁸ Sanction reports can be first (ex ante) published anonymized or not, depending on the AMF EC decision. Reports can also be anonymized ex post, following decisions of the AMF EC Chairmen (de Batz 2017a, b).

²⁹ AMF classification: insider trading, price manipulation, failure to meet with the information regulatory requirements *vis-à-vis* investors or the Regulator, failure to meet with professional obligations, proceedings, and takeovers.

³⁰ Complementary variables were built: the duration of the procedure from ignition to the sanction decision, in years, as in Karpoff et al. (2008b), and the lag between the decision and its publication, in months.
³¹ Three dummies were used to control for the impact of the anonymization: anonymized when first pub-

lished, partial anonymization, and ex post anonymization, at the AMF EC Chairmen's discretion.

³² Several variables characterize the appeals: whether the decision was appealed or not by the sanctioned entities, as in Karpoff et al. (2008b); whether the AMF appealed the decision of the AMF EC; the number of courts appealed to; whether the decision was confirmed or not; and the duration of the appeal procedure.

Table 2 Sample characteristics					
Number of observations: 52	Mean	Standard deviation	Min.	Max.	Expected impact on abnormal returns
Origin of the sanction					
Investigation (not control)	0.81	0.40	0	1	I
Breaches of insider dealing regulations	0.29	0.46	0	1	I
Price manipulations	0.10	0.30	0	1	I
Breaches of public disclosure requirements	0.63	0.63	0	2	I
Breaches of the Monetary and Financial Code and the AMF General Regulation	0.38	0.50	0	1	+
Characteristics of the sanction decision					
Cash sanction (as thousand euros) ¹	860	1354	0	8000	I
Warning	0.25	0.44	0	1	I
Blame	0.04	0.19	0	1	I
Duration of the procedure (start to sanction, as years)	2.65	1.01	1.14	5.98	I
Lag from sanction to publication (as months)	1.82	1.69	0.03	8.47	ż
Actual ² state of online anonymization	0.65	0.48	0	1	+
Partial anonymization	0.25	0.44	0	1	+
First publication anonymized	0.35	0.48	0	1	+
Top management involved in the breach(es)	0.46	0.50	0	1	I
Sanctioned individuals	0.69	0.47	0	1	ż
Public company victim of breach(es)	0.25	0.50	0	1	ż
Sanction report number of pages	11.73	5.95	3	34	I
Details of the appeals					
Appeal	0.48	0.50	0	1	+
Number of appeals	0.81	0.99	0	4	+

Table 2 (continued)					
Number of observations: 52	Mean	Standard deviation	Min.	Max.	Expected impact on abnormal returns
Rejection of the appeal	0.40	0.49	0	1	1
Appeal by AMF	0.04	0.19	0	1	I
Duration of the appeal (from the sanction, as years)	0.99	1.40	0	7.17	I
Media coverage of the sanction procedure					
Media coverage intensity before the sanction	0.06	0.05	0	0.38	I
Number of articles published between the sanction and its publication	10.5	37.0	0	248	I
Number of articles published during the week following the publication	13.8	23.0	0	114	I
Articles published in L'Agéfi or Les Échos	0.71	0.46	0	1	I
Recidivism					
Recidivism pre-AMF (up to 2003) ³	0.27	0.45	0	1	I
Recidivism post-AMF (2004-2016)	0.29	0.46	0	1	I
Stock market characteristics					
Market capitalization (on the sanction day, as million euros)	9812	15,511	8	69,393	ż
Survival to sanction (still listed)	0.88	0.32	0	1	ż
Euronext Compartment A ⁴	0.56	0.50	0	1	ż
Euronext Compartment B ⁴	0.17	0.38	0	1	ż
Euronext Compartment C ⁴	0.19	0.39	0	1	ż
Financial sector ⁵	0.38	0.49	0	1	I
Industry sector ⁵	0.15	0.36	0	1	ż
Consumer goods or services sector ⁵	0.15	0.36	0	1	ż
Technological sector ⁵	0.13	0.34	0	1	ż

Table 2 (continued)					
Number of observations: 52	Mean	Standard deviation	Min.	Max.	Expected impact on abnormal returns
Legal environment characteristics					
Year of the sanction	2009	3.4	2004	2016	I
LME 2008	0.24	0.42	0	1	I
LRBF 2010	0.40	0.49	0	1	I
President J.P. Jouyet ⁶	0.35	0.48	0	1	I
President G. Rameix ⁶	0.23	0.43	0	1	Ι
The "expected impact on abnormal returns" reflects, a priori and bas returns. For example, being investigated (<i>versus</i> controlled), reflecting returns (-). Similarly, a higher media coverage of the sanction should r decision should signal to the market that the firm pleads innocent, hen value)	cationy integration of more diversities, vision of more diversimption of more diversements, noe it could temper the	whether the variable ore severe regulator hence convert into n negative signal of t	g dates in particular should impact n y breaches, is exp nore negative abnu- he sanction decisi-	egatively $(-)$ or p egatively $(-)$ or p bected to lead to n ormal returns $(-)$. (on $(i.e.$ positive ex	ositively (+) abnormal nore negative abnormal Conversely, appealing a pected impact on stock
Sources: AMF, Factiva, Thomson Reuters. Author's calculations					
¹ Sanctions which only involved a disciplinary sanction were assigned	a zero-euro cash fine,	and sanctions involv	ing several listed	companies were a	ccounted for twice
² State of anonymization before the application of the Law on market and modernized business life, of 9, December 2016 (Law no. 2016-16)	t abuses of 21, June 20 91, IV Art. 42-46))16 (Law no. 2016-8	(19) and Law on	transparency, the	fight against corruption
³ Sanctions pronounced by the 3 Authorities which were merged to cre Conseil de Discipline de la Gestion Financière)	cate the AMF in 2003	(Conseil des March	és Financiers, Co	mmission des Opé	érations de Bourse, and
⁴ Compartment A for market capitalizations above 1 billion euros; con C for market capitalizations above below 150 million euros	mpartment B for mark	et capitalizations ab	ove between 150	million and 1 billi	on euros; compartment

⁶J.P. Jouyet from December 2008 to July 2012 and G. Rameix from August 2012 to July 2017

⁵According to Euronext classification

victims of the financial misconduct being sanctioned); (3) the attributes of the defendant (such as the moral form, whether an individual (employee, manager, other) was sanctioned, the top management involvement (chairman, CEO, CFO, owner of the company),³³ the survival of the firm to the sanction, recidivism before and/or after the AMF creation,³⁴ place of listing, stock market capitalization, business sector³⁵); (4) the media coverage of the sanction (media exposure intensity before the sanction, the number of articles published between the decision and the publication and over the week following the publication, articles on the sanction published in top tier journals, *L'Agéfi* and *Les Échos*); and (5) some time and legal indicators (AMF chairmen of the board, financial regulations, real GDP growth rate). A comprehensive correlation analysis was carried the dataset.³⁶

The sample was restricted to sanctioned listed companies, which were historically the most frequently sanctioned population (42% of the sanctions), followed by asset management firms. The initial sample covered 134 cases, in which 129 sanctions impacted 105 companies. Some sanctions involve several listed companies. Some firms were repeatedly sanctioned (*i.e.* recidivism), when taking into account branches of groups. These repeat offenders were sanctioned on average three times, ranging from two up to nine sanctions. They were most frequently financial institutions.

The final sample covers less than half of the initial sample: 52 sanctions (*i.e.* on average 4 sanctions per year) against 40 daily listed companies. 6 of them are no longer listed, following mergers and acquisitions (M&As) or bankruptcies. The sample is comprised of all the firms which were daily listed on the Paris stock-markets,³⁷ from the 120 trading days before ignition of the procedure, until 120 trading days after the publication of the sanction (*i.e.* daily listed on average over 3.4 years). The reasons for exclusion from the sample include: early delisting, late listing, temporary

³³ Generally speaking, a focus is made on the individuals within an organization convicted of crime, as recommended by Cohen (1996), either employees (with a principal-agent relationship derived from the employer-employee contract) or top managers. From an investor's perspective, the top management involvement in a fraud could be particularly detrimental, demonstrating the improper management of the firm and questioning the capacity of the management to handle future challenges. Karpoff et al. (2008b) showed how financial mis-presentation can prejudice careers of top managers: more than 90% of them lose their jobs by the end of the US SEC enforcement procedure.

³⁴ Recidivism is one of key aggravating factor regularly stated by the AMF to define the size of the sanction. Repeat offenders are sanctioned more severely than first-time offenders.

³⁵ According to the Euronext classification. The most frequent sectors (dummy variables) are: financial services, industry, consumer goods and services, and technology.

³⁶ Detailed results and analyses are available on demand or in de Batz (2018).

³⁷ Euronext is organized around three pillars:

¹⁾ The European Union regulated market for equity securities operates in five markets (including Paris). They are segmented by market capitalizations: compartment A (above 1 billion euros), compartment B (from 150 million to 1 billion euros), and compartment C (below 150 million euros).

Alternext targets small-and-mid-sized companies by offering a simplified access to capital markets with fewer requirements and less stringent ongoing obligations than on the EU-regulated market.
 The free market provides the easiest access to capital markets through a direct quotation procedure for any company, whatever the size (from micro-cap to medium-sized international companies) searching to access capital markets (free from the Euronext's eligibility criteria and information disclosure requirements). This market targets primarily sophisticated or professional investors.

suspension, or lower-than-daily quotation frequency.³⁸ Such companies could be already ailing, experiencing financial difficulties (announcing a delisting or a failure in the near future), less traded (hence less liquid, questioning the price formation mechanism around the events), or undergo exceptional events justifying a temporary suspension (such as M&As). All these reasons are likely to interfere with the event and to bias (to the down- or up-side) the market responses to the news of a sanction. Additionally, four sanctions on a multinational bank daily traded in Euronext Paris were excluded, due to the limited share of activities in France.³⁹ Acquittal decisions (11 cases) were also removed from the initial sample, given the different nature of the verdict. To avoid overlap and enable data clustering, two concomitant sanction procedures were merged (cash fines and disciplinary sanctions), to assess the cumulated severity of the regulatory decisions. Finally, five sanctions were dropped due to major confounding events, such as the outcome of a major lawsuit, the start of a safeguard procedure, or changes of name. The risk of introduction of biases through the sample selection is tamed by the comprehensiveness of the sample of sanctioned listed companies. Complementarily, some initially excluded sanctions were included in complementary robustness analyses, as well as settlements with listed companies.

Table 3 compares the descriptive statistics of the initial and the final samples of sanctioned listed companies. For the final sample, sanctions massively followed investigations (81% in the sample), which target the most serious regulatory breaches (*i.e.* the three market abuses). 1.4 regulatory breaches were committed per sanction on average, the most frequent being dissemination of false information (63%), failures to meet with professional obligations (38%), and insider trading (29%). The great majority of companies were large, as 56% of them were listed on the Compartment A and 17% on the Compartment B of Euronext. The average market capitalization amounted to 9.8 billion euros (on the day preceding the sanction), ranging from 8 million up to 69 billion euros, with a standard deviation of 15.5 billion euros. 48% of the decisions were appealed, with an 84% confirmation rate of the AMF EC's decision.

Most of the divergences between the sample and the average for listed sanctioned companies derive from the higher share of financial companies in the sample (38%, against 25%). These sanctions targeted top tier universal banks, with higher-thanaverage market capitalizations (by 42%),⁴⁰ and a lower likelihood of bankruptcy (the Central Bank being the lender of last resort). The gap in market capitalizations can also be explained by the fact that smaller companies are more frequently not

³⁸ In our sample, delisting can be accounted for three main sets of reasons, by decreasing frequency: (1) bankruptcy; (2) mergers or acquisitions with/by another listed company, leading to delisting; and (3) managerial decision to delist due to the regulatory constraints and the legal and financial risks associated, preferring other financing sources (less regulatory constrained). Karpoff et al. (2008a) also found for the US that there is high delisting rate, which reduces massively the size of the sample. The study also stresses that the delisted companies tend to be associated the poorest stock performance over the whole enforcement period.

³⁹ Given the size of the bank (14 times the average market capitalization of the sample), and its listing on several stock exchanges, any action from the French AMF would unlikely provoke a significant abnormal reaction from global shareholders. Additionally, confounding events could lead to misinterpret the results.

⁴⁰ When excluding the 4 sanctions of the major international bank excluded from the sample.

	All listed companies ³	Sample listed com- panies
Number of sanctions	129	52
Number of sanctioned companies	105	40
Of which bankrupted	23 (19 before sanction)	2
Investigations (as % of total)	88	81
Number of regulatory breaches per sanction	1.5	1.4
Main activity sectors:		
Financials (as % of total)	25	38
Consumer goods or services (as % of total)	22	15
Industrials (as % of total)	22	15
Technology (as % of total)	13	13
Average cash fine ¹ (as thousand euros)	693	882^{4}
Average duration of procedure (as years)	2.7	2.6
Average market capitalization (as billion euros, on the	11.9	9.8
day preceding the sanction) ²	$(6.9)^5$	

 Table 3 Characteristics of the sanctions of listed companies versus the sample

This table presents the main features of the initial and the final samples of listed companies being sanctioned by the AMF from 2004 to 2016. The final sample ("Sample Listed Companies") is comprised of the decisions involving a daily listed firm, from 120 trading days before the ignition of the enforcement procedure, until 120 days after the publication of the decision. Major confounding events or duplicated and acquittal procedures were also excluded as well as abnormally big firms

Sources: AMF, Thomson Reuters. Author's calculations

¹Excluding acquittals and counting only one time sanctions involving several listed companies

²For companies still listed when being sanctioned guilty, on the day of the sanction decision

³Listed companies cover all the sanctions of listed companies, including acquittals

⁴Excluding the sanctions with only a disciplinary sanction (meaning a null cash fine)

⁵Average market capitalization when excluding the 4 sanctions on the major international bank excluded from the sample

daily quoted (hence excluded from the sample), or experience financial difficulties. It can lead to quotation suspension or bankruptcy shortly after the sanction, which adds to the reasons for being excluded from the sample. Financial firms are also the most likely to reoffend (de Batz 2017a, b). All in all, the sample cash fines are 28% higher-than-average, as recidivism and size are two of the few regulatory determinants of cash fines.

5 Impact of sanctions on listed companies

5.1 Impact on stock returns

Four event studies are conducted for each step of the enforcement (see Fig. 1). Step 1 is the beginning of the (AMF internal) procedure, with the approval of an investigation (for alleged market abuses) or a control (for breaches to professional obligations). Step 2 is the statement of objection, when the incriminated firm is notified by the Board of the AMF that it is being investigated for characterized regulatory breach(es) and asked for additional information. Given these elements, the Board may transfer the case to the AMF EC, initiating the "judicial part" of the procedure. Step 3 is the sanction decision made after the AMF EC hearing (*i.e.* the trial), followed by the (possibly anonymized) publication of the sanction report on the AMF website (step 4). Since 2010, the AMF EC hearings have been opened to the public, without naming ex ante the case(s) under review. Top-tier financial journalists typically attend them. Hence, newspaper articles can be written over the 50-trading-day average lag between the decision and its publication (42% of the sample). The echo of sanctions in the press is even greater after the publication (85% of the sample). Hence, returns can be expected to react to the anticipated publication of the sanction.

As described in the methodology, for every step of the procedure, "abnormal" returns are calculated over the event window [-10; +120] with respect to the event in t = 0 from the "normal" parameters estimated over the estimation window [-120; -11] (see specification (1)).⁴¹ Including days before the event investigates for anticipation following leakages or speculation. Abnormal returns are then cumulated along time and averaged across sanctions to draw some inferences on the abnormal reactions following the milestones of the proceeding (see specification (2)). Figure 2a–c and Table 4 report, for every step of the procedure, Average Abnormal Returns (AAR_t), and Cumulative Average Abnormal Returns ($CAAR_{[t_1;t_2]}$) for the whole sample (n = 52). Complementarily, a set of cumulative abnormal returns for sanctioned firms and for every step is presented in Fig. 3a–e. They demonstrate adverse and genuine effects of some steps of the sanction procedure on returns of sanctioned listed firms over the period under review.

On the one hand, shareholders do not react significantly to the early steps of the procedure: no significant abnormal return follows either the beginning of the procedure, or the statement of objection. We fail to reject the hypothesis H_1 , in line with expectations. This result for the beginning of the procedure rejects any breach of confidentiality from part of the AMF teams in charge of internal procedures. Leakages to market players or use of insider information could have caused a reaction in stock returns. Secondly, the absence of abnormal returns following the statement of objection demonstrates the lack of insider trading from part of the AMF and of the investigated company, after learning about a procedure that can end up with a sanction.

On the other hand, as expected (see hypothesis H_2), the last two steps of the procedure trigger statistically significant reactions: shareholders suffer abnormal losses following the sanction decision, and its publication. The regulatory decision on the guiltiness of a given listed firms is negative information to the market: on average, returns contract by a cumulated abnormal 0.9% over the period [-1;+3] in event

⁴¹ Robustness checks were performed by modifying the length of estimation window (100 and 90 days). The results are not sensitive to such variations. Detailed results are available on demand.









Fig. 2 Average abnormal returns and cumulative average abnormal returns for the different milestones of the sanction procedures. Abnormal returns are computed given the market model parameters estimated with OLS with White-corrected standard errors, through the period [-120; -11] in event time (see specification (1)). Event time is days relative to the step of the sanction procedure under review. The sample is composed of 52 sanctions of daily-listed companies over the period 2004–2016. Average abnormal returns AAR_t and *CAAR*_[*t*_1:*t*_2] are calculated using the specification (2). *Sources*: AMF, Thomson Reuters. Author's calculations

time (significant at the 10% level). Additionally, they lose 0.8% over the period [-1; 0] following the publication of the decision (significant at the 1% level) and 1.1% over the period [-1;+3] (significant at the 10% level). Interestingly, there is some anticipation in the reaction, before the publication, as in previous studies: abnormal returns turn significantly negative, possibly anticipating the outcome of the decisions and newspaper article, or due to leakages of information to insiders, as seen in other jurisdictions.⁴² 62% of the guilty companies exhibit negative abnormal returns on the day of the publication of the sanction, ranging from -5.3 to +5.0% (with a 1.5% standard deviation). Three days after the publication, 63% of the companies suffer cumulated losses, ranging from -12.1% to gains of +7.8% (4.4% standard deviation). The contraction peaks 6 days after the publication (*CAAR*_[-1:+6] = -1.3%,

(b) Cumulative Average Abnormal Returns

⁴² For example, ordered chronologically: Pritchard and Ferris 2001, Griffin et al. 2004, Djama 2008, Dyck et al. 2010, Haslem et al. 2017 and Armour et al. 2017.

	Beginning of cedure (contr investigation)	pro- ol or	Statement of	objection	Enforcement mittee and sa decision	com- nction	Publication o sanction deci	f the sion
t	$\overline{\text{CAAR}_{t}(\%)}$	t-stat	$\overline{\text{CAAR}_{t}(\%)}$	t-stat	$\overline{\text{CAAR}_{t}\left(\% ight)}$	t-stat	$\overline{\text{CAAR}_{t}(\%)}$	t-stat
-1	0.4	0.8	0.7**	2.2	-0.4	-1.2	-0.5***	-2.7
0	1.4	0.7	0.8	1.5	-0.6*	-1.7	-0.8***	-3.0
1	2.0	1.0	0.4	0.5	-0.6	-1.7	-0.7*	-2.0
2	1.5	0.7	-0.1	-0.1	-0.9*	-1.9	-0.9*	-1.9
3	1.9	0.9	0.4	0.5	-0.9*	-1.8	-1.1*	-1.7
4	2.0	1.0	0.0	0.0	-0.8	-1.3	-0.9	-1.5
5	2.6	1.3	0.0	0.0	-0.8	-1.2	-1.1*	-1.8
6	3.3	1.6	-0.2	-0.2	-1.3	-1.6	-1.3**	-2.2
7	2.6	1.2	-0.2	-0.3	-1.3	-1.5	-1.3*	-1.8
8	1.0	0.4	-0.3	-0.4	-1.4	-1.7	-0.9	-1.2
9	1.2	0.5	-0.4	-0.4	-1.1	-1.3	-0.8	-0.9
10	0.8	0.3	0.0	0.0	-1.0	-1.1	-1.5	-1.6
20	3.9	1.4	-0.7	-0.4	-1.3	-1.0	-0.8	-0.6
40	7.4	1.5	-1.8	-0.7	-1.9	-1.0	1.2	0.6
60	9.1*	1.7	-0.2	-0.1	-3.7	-1.4	0.2	0.1
120	11.8*	1.7	3.6	0.6	-7.1	-1.5	-6.6	-1.5

 Table 4
 Cumulative average abnormal returns following the sample of 52 sanctions of the listed companies

This table reports the cumulative average abnormal returns (CAAR_t) up to a specified day *t* in event time for the four main steps of the sanction procedure, as defined in specification (2). Event time is days relative to the step of the sanction procedure being analyzed and t = 0 is the event itself. Abnormal returns are computed given the augmented market model parameters, which are estimated with OLS through the period [-120; -11] in event time. The sample is composed of the 52 companies which were sanctioned guilty by the AMF from 2004 to 2016 and were daily quoted all through the sanction procedure

Sources: AMF, Thomson Reuters. Author's calculations

*, **, and *** denote statistical significance at the 10%, 5%, and 1% level

significant at the 5% level). In the longer run, cumulative average abnormal returns following the sanction decision remain negative though statistically insignificantly $(CAAR_{[-1;+60]} = -3.7\%)$. This higher contraction echoes the lag between the AMF EC hearing and the publication of the decision: 50 trading days on average in the sample (de Batz 2017a). Hence, the cumulated contraction 60 days after the sanction would incorporate the compounded abnormal reactions to the sanction and to its publication, estimated over a window excluding the sanction decision and its publication.

Given the limited (though exhaustive) number of observations, and to ensure that the presence of outliers does not bias the results, two complementary robustness checks were conducted. A bootstrapped analysis of the robustness of standard errors was conducted 1000 times, with a confidence interval of 95%. Complementarily, abnormal returns were winsorized before estimating the test statistics, as in Armour



Fig. 3 Cumulative average abnormal returns and cumulative abnormal returns for some sanction procedures. Abnormal returns are computed given the market model parameters estimated with OLS with White-corrected standard errors, through the period [-120; -11] in event time, using specification (1). They are then cumulated along time. The figures depict abnormal price developments (on average or specific sanctioned firms), rebased 100 on the day of the event. Event time is days relative to the step of the sanction procedure under review. The sample is composed of 52 sanctions of daily-listed companies over the period 2004–2016. Sanctions are numbered chronologically. *Sources*: AMF, Thomson Reuters. Author's calculations

et al. (2017): all abnormal returns outliers to a 90th percentile were excluded from the data, meaning that all data below the 5th percentile are set to the 5th percentile, and data above the 95th percentile are set to the 95th percentile. The magnitudes of the *CAAR* were confirmed and turned out to be slightly more significant and persistent in time with winsorized abnormal returns.⁴³

All in all, in the short run, these sequential event studies confirm the reactions observed in previous research. They also contribute to improving the quality of the

⁴³ Detailed results are available on demand.

assessment of the spillovers of sanctions in France. In fact, the exhaustive sample of daily listed guilty companies implies a broader scope of analysis and a higher granularity. No abnormal reaction was measured through the early stages of the enforcement procedure, rejecting breaches of confidentiality either by the Regulator and the defendant(s). Subsequently, the results are coherent with past studies on the French sanctions, though to a lower extent: sanction decisions and their publications convey information and impact negatively returns of listed companies in the short run.⁴⁴ As in the literature, some anticipation in the outcome was measured, with the negative correction in prices. Additionally, contrary to the efficient market hypothesis, investors' reactions tend to be scaled in time: the spillovers of sanctions on the stock returns take some time to fully materialize. Some investors will react immediately after learning the sanction. Conversely, various reasons can contribute to this inefficiency of financial markets, leading to no or postponed reactions: the time to access information (initially unaware, herd behaviors), the poor financial education (misunderstanding of regulatory breaches), or the avoidance of financial consequences (fees due to portfolio rebalancing, deterring fiscal consequences, no investment alternative, etc.).

In the longer run, past literature estimates a large range of impacts, from positive⁴⁵ to very negative.⁴⁶ Some studies conclude that a fraud durably affects returns, up to three years after the news, when using lower frequency data (Leng et al. 2011; Dyck et al. 2010). Such estimates must be taken with caution as the further from the event, the more likely confounding events will interfere with abnormal returns. The impact of French sanctions on guilty daily-listed companies in the longer run remains limited compared with international estimates. Our results demonstrate that, over the six months following the sanction (either decision or publication), *CAAR_t* remain negative, even though they are not significantly different from zero. That could be explained by the higher volatility in the long run. Finally, it is likely that the reaction following the sanction decision is partly confounded with the reaction following the magnitude of average abnormal returns becomes more substantial: -3 to -4% cumulated losses 60 trading days after the sanction, estimated over an estimation window excluding any event related to the sanction.

⁴⁴ Kirat and Rezaee (2019) concluded with a statistically significant -0.7% in AAR_0 on the day of the announcement of the sanction in the press and a $CAAR_{[-2:2]}$ of -1.7%, with a sample of 47 companies over the period 2004-2017. Djama (2008) found no impact of the beginning of the procedure and a significant negative impact of the publication of the decision (-6.9% in $AAR_0 - 8.3\%$ in $CAAR_{[0:1]}$), for accounting fraud with a sample of 37 sanctions of 28 listed companies, from 1995 to 2005.

 $^{^{45}}$ Such as +2.96% in one-year stock performance following a 1-standard deviation increase in the financial penalty for 20 country panel, in Köster and Pelster (2017).

⁴⁶ The maximum was -34.4% in the US cumulated over the days for which the firms were subject to a regulatory event, in Karpoff et al. (2008a).

⁴⁷ As stated by Armour et al. (2017), multi-stage events make it difficult to ensure that the later stages really relate to the original announcement and not to further information that was released during subsequent stages or conversely that relevant information was not released between the reported stages.

5.2 Complementary results and information content of the sanction decisions

Robustness checks with larger or sub-samples and complementary analysis were conducted to test the sensitivity of the results to the hypotheses. Results are robust and complementary with conclusions previously described.⁴⁸ They show that the French enforcement actions are not trivial.

Firstly, including into the initial sample the four sanctions pronounced against a major international bank leads to similar impacts with the central event study. Still, abnormal returns are lower and less significant, supporting the hypotheses which led to their initial exclusion.

Secondly, the impact of the Great Financial Crisis was tested, given its magnitude and spillovers. Financial firms, which were at its origin and its main victims, are the most frequently sanctioned listed companies. The crisis also led to a tightening of financial regulation and supervision, in particular regarding sanction powers at the European and French levels (de Batz 2017b). The sample was split by publication dates, either before or after the crisis.⁴⁹ The re-estimated event studies show that the information content of the publication of sanctions seems to have increased since June 2007, with significant and more negative abnormal returns. They suggest a reinforced market awareness and risk sensitiveness to sanctions and regulatory interventions.

Thirdly, following past literature, offenses were sorted into two main categories: whether they hit related parties or not (*i.e.* third parties), based on the AMF split of regulatory breaches.²⁹ The event studies demonstrate that the three breaches impacting related parties lead to higher abnormal negative returns after the publication (by declining order of magnitude and significance): (1) insider trading; (2) dissemination of false information (*vis-à-vis* investors or the Regulator); and (3) not complying with one's professional obligations. Conversely, price manipulation (*i.e.* hitting third parties) does not cast significant abnormal returns (though with a limited sample size). Such results confirm past studies as investors tend to react more when they are impacted by the financial misconduct (*i.e.* by being a related party).

Fourthly, the event studies were re-estimated for two subsamples based on the "seriousness" of the verdict,⁵⁰ capitalizing on the guidelines given by the AMF on how to set the sanction.¹⁸ The results demonstrate that the mere cash fine is uncorrelated to the magnitude of abnormal returns. That may be explained by low level of

⁴⁸ Detailed results are available on demand or in de Batz (2018).

⁴⁹ As in the literature, two alternative starting dates were tested: June 2007 (Armour et al. 2017), with the beginning of the US subprime crisis, or September 2008, with Lehman Brothers' bankruptcy (Kirat and Rezaee 2019).

⁵⁰ First subsample "Average" (19 sanctions): cash fines higher than the average.

Second subsample "3 factors" (19 sanctions) if two out of the three following conditions: a cash fine higher than the median; a disciplinary sanction (warning or blame); and recidivism (pre- and/or post-AMF creation).

average cash fines (in absolute and relative terms).⁵¹ Some cumulated dimensions of the seriousness of the decision (subsample "3 factors")⁵⁰ point to a more severe financial misconduct, leading to more negative abnormal returns. That confirms the initial hypothesis H_4 that not only will the mere fact of being sanctioned be priced in abnormal returns, but the nature of the sanction will also negatively influence the results.

Fifthly, no significant abnormal returns followed the 7 decisions for which the identities of the sanctioned firms were anonymized when first published, whatever the step of the procedure. This confirms the lack of breaches to confidentiality through the whole enforcement process.

Finally, two complementary samples of decisions were subjected to the event study methodology: (1) the 11 acquittals of listed companies, and (2) the 5 settlement decisions involving subsidiaries of daily listed companies (three French financial groups).⁵² For both samples, the expected information content of the decisions was not straightforward, under the assumption that sanctions convey information to investors. Firstly, acquittals can mean innocence and no fine has to be paid (positive signal). Conversely, they signal serious doubts from the Regulator regarding a firm (negative signal), as only the most severe alleged financial wrongdoings are brought to the AMF EC. Other breaches are dealt with confidentially and bilaterally, between the AMF and the regulated entity. Additionally, acquittals frequently result from procedural irregularities, or prescription limit of the incriminated regulatory breach(es), which do not exonerate the entity from any liability. Secondly, settlements are alternative lighter procedures, dedicated to less serious regulatory breaches than sanctions. They do not imply guilt recognition from part of the defendant. Still, they result from significant financial misconducts. Under the rationality of investors and efficient market hypotheses, abnormal returns following settlements are expected to be negative, though lower than for the sample of guilty sanctions.

Acquittal decisions convey mixed signals: positive statistically significant abnormal returns on the day of the sanction $(AAR_0 = +1.1\%)$, at the 10% level), followed by a negative one after the publication $(CAAR_{[0;+3]} = -3.1\%)$, at the 10% level). The results could be explained by the limited sample size. They also echo the divergent conclusions on the impacts of allegation of financial misconduct (*i.e.* the mere fact of being investigated) as well as on acquittal decisions, in other jurisdictions.

For the small sample of settlements, no significant abnormal returns followed any step of the proceeding. The markets do not price in the additional information on the firm's compliance with regulation. Similarly, Haslem et al. (2017) found market reactions to settlements being the least negative and negligible, whatever the outcome. The lack of reaction to settlements questions the information content of such procedures, and the credibility of the AMF communication and decisions *vis-à-vis* investors.

⁵¹ Compared to the market capitalizations, to fines by other French Regulatory Authorities, or to international standards. The average abnormal market correction stands for a limited 0.001% of the market capitalization.

⁵² 32 settlements were signed from 2012 to 2016. A settlement is an alternative and shorter kind of sanction dedicated to the least severe regulatory breaches (until late 2016), subject to an AMF proposal and an acceptance by the firm.

5.3 Impact on market values and reputational penalty, following the publication of the sanction

This section investigates the hypothesis H_3 : does a reputational penalty explain part of the abnormal returns following the sanctions of listed firms? The focus is limited to the step of the procedure triggering the biggest and most significant reactions in *CAAR*: the publication of the sanction. From specification (3), the impact on market capitalizations $SL_{[-1;+t]}$ is estimated from the $CAR_{[-1;+t]}$, from the day preceding the event until *t* days after in event time. On average, sanctioned firms lost in equity $ASL_{[-1;0]} = -45,200$ euros, $ASL_{[-1;+1]} = -74,600$ euros, and $ASL_{[-1;+6]} = -32,000$ euros respectively. There is a wide range of reactions, suggesting that not all frauds are equally important to shareholders. For example, over the event window [-1;+1], $SL_{[-1;+1]}$ range from losses of 2.2 million euros (-6%loss in value) up to gains of 870,900 euros (+2.7% in value), with a standard deviation of 363,000 euros.

Hence, markets do integrate the information of the sanction sent by the Regulator as a negative signal, but only to a limited extent (see Table 5). The impact on the market capitalizations is small in absolute, as well as in relative terms. On average, for the sample, the cash fine is 12 times bigger than the abnormal market correction. Put it differently, on average, the reputational impact for being sanctioned *RL* is positive, after subtracting the cash fine from the estimated abnormal market reaction (see specification (4)). One gains from being sanctioned. Still, the dispersion is very high. Market efficiency, in that sense, is limited. This result questions the credibility of AMF's sanctions. We fail to reject the null hypothesis H_3 that there is no reputational penalty subsequent to the publication of a regulatory sanction. It is all the more striking that the regulatory fines are already perceived as low compared to the legal authorized maximums (100-million-euro threshold for any professional under the AMF supervision, see Table 1), as well as in relative terms (standing for 0.01% of the market capitalization on average) and in international standards.⁵³

Sanctioned financial misconduct implies statistically significant market penalties but no reputational penalty. One would expect that such regulatory decisions impact negatively the perception by shareholders, stakeholders, etc., as it is the case in other jurisdictions (US and UK in particular). The "reputational gain" for French listed firms from being sanctioned questions the severity of the verdict (in particular the levied financial penalties) and, more broadly, the credibility of the Regulator. In sharp contrast, in the US, the reputational penalty represents 90% of the equity loss, according to Karpoff and Lott (1993). In Karpoff et al. (2008a), the reputational penalty for cooking the books in the US stands for 7.5 times all penalties. Firms lose 38% of their market values when their misconduct is reported, 2/3 of which being a

 $^{^{53}}$ In the US, from 1978 to 2002, the average monetary fines stood for 107 million dollars. In the US, in Karpoff and Lott (1993), the fines stand for 6.5% of the stock losses. In the UK, based on the estimations of Armour et al. (2017), the average fine set by the Regulator amounts to 0.26% of the market capitalization.

	[-1;0]	[-1;+1]	[-1;+6]
$CAAR_{[-1;t]}$, as %	-0.8%***	-0.7%*	-1.3%**
$ASL_{[-1;t]}$, as euros	-45,200	-74,600	- 32,000
Standard error, as euros	192,000	362,910	616,843
Minimum $SL_{[-1;t]}$, as euros	-834,400	-2,236,000	-3,533,300
Maximum $SL_{[-1;t]}$, as euros	+192,118	+870,900	+1,192,700
Negative $SL_{[-1:t]}$, as % of total	60%	56%	58%
ASL, as % of average cash fines	5.1%	8.5%	3.6%

Table 5 Average stock losses/gains following the publication of the sanctions

The cumulative abnormal returns from t_1 to t_2 (CAR_[t_1:t_2]) used to estimate the impact of sanctions are estimated after the publication of the sanction (see specification (2)), for the sample of 52 listed companies over three different event windows. The subsequent Average Stock Losses/Gains ($ASL_{[t_1:t_2]}$) are calculated with a residual approach (see specification (3)), by deducting the fines from the estimated abnormal impacts (in euros)

Sources: AMF, Thomson Reuters. Author's calculations

*, **, and *** denote statistical significance at the 10%, 5%, and 1% level

pure reputational penalty. In the UK, the reputational penalty exceeds by 9 times the cash fine, according to Armour et al. (2017).

5.4 Cross-sectional determinants of the stock market's reaction

In this section, we use a cross-sectional regression to explain the determinants of the abnormal returns incurred by each sanctioned firm in the aftermath of the publication of the sanction (*i.e.* the most significant results). The dependent variables are the $CAR_{i,[-1;+t]}$, as defined in specification (2). For a cross-section of sanctions, we run OLS regressions with robust White-corrected standard errors of $CAR_{i,[-1;+t]}$ against all the explanatory variables from the dataset (firm, sanction, and environment characteristics, see Table 3), based on specification (5). The results for three following models are presented, robust with the exhaustive cross-sectional test.

Model 1 is estimated from the day preceding the publication until *t* days (t = 0 and + 6), for each sanction *i*, with the following explanatory variables: a dummy for sanctions following an investigation (*i.e.* the most serious regulatory breaches); a variable for the length of the procedure from the beginning until the sanction decision (synonym of complexity); a dummy for sanctions anonymized by the AMF; a dummy for the rejection or dismissal of the appeal (confirmation of the AMF verdict); a dummy for the media attention, when articles are published following the sanction publication of the top-tier financial journals (*L'Agéfi* or *Les Échos*); a dummy for firms which "survived" the sanction (*i.e.* are still listed); a dummy for the largest firms (*i.e.* listed on the Euronext Compartment A); a dummy for Euronext industrial firms; a dummy for Euronext technological firms; and the real quarterly French GDP year-on-year growth rate when the sanction was published (synonym of the economic conditions);

Two alternative models (models 2 and 3) were estimated for the peak in significant cumulative abnormal returns ($CAR_{i,[-1:+6]}$), with the following alternative

variables: a dummy if the top management of the firm was involved in the regulatory breach(es) (chairman, CEO, CFO, owner of the firm); a media intensity variable before the sanction (ratio of articles mentioning the firm over the 20 days preceding the sanction to the number over the preceding year); a variable for the number of articles mentioning the sanction published over the week following the publication; a dummy for Euronext consumer goods or services firms; a dummy for Euronext financial firms; and a dummy for the sanctions published under the financial law LME (2008 to 2010).⁵⁴

Table 6 reports the strongly robust results. The fits of the models over the period [-1;+6] are particularly interesting given their robustness, and the fact that more time is given to market players to react to the news of the sanction (*i.e.* inefficient markets). The following takeaways can be drawn regarding the information content of sanctions and their interpretation by the market.

Four aspects of the sanction contribute to significantly more negative abnormal returns: being investigated (*versus* controlled), longer procedure, the top management's involvement in the regulatory breach(es), and a higher media coverage. Interestingly, negative abnormal returns appear higher in better economic times, possibly as stronger forces than sanctions may play during an economic crisis and lead to global negative trends.

Regarding the sanctioned firms, size (being listed in the Euronext Compartment A, which is also positively correlated with recidivism) curbs negative abnormal returns. In terms of sectors, as expected, model 3 shows that financial firms will endure more negative abnormal returns for being sanctioned. This sector is the most frequently sanctioned, and the most prone to recidivism. Being a technological firm will also contribute to more negative abnormal returns, conversely to being a consumer goods or services firm.⁵⁵

Surprisingly, appealing a decision, which could stand for a positive signal (claiming for one's innocence), sends mixed and limitedly significant information: positive in the very short run, before turning negative, at the 10% level. It could be accounted for by the historically low probability of success of appeals. Additionally, anonymization leads to significantly more negative abnormal returns, though only in the short run. The consecutive regulatory tightenings did not impact significantly returns.

The last two takeaways can be drawn from insignificant variables. Firstly, the three variables controlling for the most straightforward features of the sanction decision (cash fine, warning, and blame) do not statistically significantly influence market reactions.⁵⁶ The fine and disciplinary sanctions do not serve as signals of the seriousness of the misconduct. Nor do the regulatory breaches committed by the

⁵⁴ Loi de Modernisation de l'Économie (LME, no. 2008-776).

 $^{^{55}}$ In the case of the industrial sector, the estimations of model 1 over the periods [-1; 0] and [-1; +6] show that the negative contribution is due to some market anticipation, which is more than compensated in the subsequent period.

 $^{^{56}}$ Derived variables were also tested, as in Armour et al. (2017), such as the natural log of the cash fine or the ratio of the fine to the market capitalization the day before the sanction. The results were also insignificant.

Table 6 Determinants of cumulative abnormal	returns following	g the publicati	on of sanction de	cisions: cross-s	sectional regression	IS		
	$CAR_{[-1;0]}$		$CAR_{[-1;+6]}$					
	Model 1		Model 1		Model 2		Model 3	
	Coef.	RSE ¹	Coef.	RSE ¹	Coef.	RSE ¹	Coef.	RSE ¹
Constant	4.87***	(1.29)	11.71^{***}	(2.38)	12.29***	(2.19)	13.29***	(2.37)
Origin of the sanction								
Investigation (not control)	-1.31*	(0.73)	-3.51**	(1.44)	-5.13^{***}	(1.38)	-3.74^{***}	(1.25)
Characteristics of the sanction decision								
Duration procedure	-0.73^{***}	(0.24)	- 2.05***	(0.39)	-1.95***	(0.37)	-2.09***	(0.44)
First publication anonymized	-1.45**	(0.63)	- 1.02	(1.00)				
Top management involved					-2.06^{**}	(0.93)	-2.45**	(66.0)
Appeals & Media								
Reject of appeal or withdrawal	0.97*	(0.53)	-1.75*	(06.0)				
Media coverage intensity before sanction							-11.20^{***}	(4.07)
Number of articles published over the week after pub.					-0.046***	(0.02)		
Article(s) in L'Agéfi or Les Échos	-1.25**	(0.60)	0.74	(0.92)				
Stock market characteristics								
Survival to sanction (still listed)	-0.013	(0.63)	-5.62***	(1.20)	-5.30***	(0.78)	-5.84***	(1.25)
Euronext Compartment A	-0.20	(0.61)	3.90^{***}	(0.91)	3.99***	(0.98)	5.46***	(86.0)
Industrial sector	-1.16*	(0.66)	3.24**	(1.30)	4.54***	(1.25)		
Technological sector	-2.60^{***}	(0.55)	- 0.94	(1.34)				
Cons. goods or serv. sector					5.79***	(1.29)		
Financial sector							-3.33^{**}	(1.28)
Legal environment characteristics								
Real YoY growth rate	-0.49^{***}	(0.14)	-0.64^{***}	(0.22)	-0.71^{***}	(0.23)		
LME law (2008–2010)	-2.20^{***}	(0.75)	- 4.69***	(1.52)	-4.49***	(1.44)	-3.79***	(1.19)

330

	$CAR_{[-1;0]}$		$CAR_{[-1;+6]}$					
	Model 1		Model 1		Model 2		Model 3	
	Coef.	RSE ¹	Coef.	RSE^{1}	Coef.	RSE ¹	Coef.	RSE ¹
7	52		52		52		52	
22	0.479		0.695		0.752		0.638	
<pre>test Prob > F</pre>	0.415		0.309		0.437		0.442	
This table reports results from OLS r	egressions (using White	e-corrected star	ndard errors) from	specifications	(5) for the model	1 (over the ev	vent windows [-1: 4	-01 and

[-1; +6]), and the models 2 and 3 (over the event window [-1; +6]). The dependent variables are the Cumulative Abnormal Returns from one day before the publication of the sanction decision (t = -1) until t days after: $CAR_{i_1(-1;+1)}$, for $i = 1, \dots, 52$ and t = 0 and + 6. Abnormal returns are computed using the augmented market model. The sample is composed of the 52 companies which were sentenced guilty by the AMF from 2004 to 2016 and were daily quoted through the whole sanction procedure. A negative coefficient means a reduction in abnormal returns (i.e. a higher loss or a lower gain), and conversely for a positive coefficient

Sources: AMF, Thomson Reuters. Author's calculations

RSE: White-Robust Standard Errors

*, ** and *** denote statistical significance at the 10%, 5%, and 1% level

sanctioned company, and recidivism (either before or after the creation of the AMF), despite being key parameters considered by the Enforcement Committee to set its verdict. This can be partly accounted for by the fact that the survival to sanctions of listed firms (*i.e.* still being listed) is significantly negatively correlated with abnormal returns. Secondly, the consecutive Chairmen of the AMF, named by Government decree, and the EC AMF chairmen do not appear to have influenced the information content of the sanctions, as perceived by market players. This supports the regulatory independence of the regulatory actions.

5.5 Discussion of results

The event studies and the cross-sectional regressions demonstrate that, over the period under review, the markets do price in the information of the sanction, but to a moderate extent. The results on CAAR are limited but consistent with most past studies. They are supported by a precise unique identification of the announcement date and the exhaustivity of the data set. Additionally, they show that the most classical seriousness determinants of sanctions were hardly taken into account by the market: cash fines, disciplinary sanctions, regulatory breaches being sanctioned, and recidivism. Still, some complementary signs of seriousness are incorporated into prices, such as being investigated (not controlled), longer procedures, or the involvement of the top management of the firm. It may be due to the fact that the fines set by the Regulator and echoed by the market (i.e. "reputational penalty") are limited in absolute and relative terms, when compared with the legal maximums, with other French regulatory authorities, or with other jurisdictions. Indeed, in the US, the use of financial fines is less common than in France (8% of the sample in Karpoff et al. 2008a), but the amounts are much more significant (average of 107 million dollars, in Karpoff et al. 2008a). That could plead for scarcer and much more severe sanctions in France, though Armour et al. (2017) concluded, for the UK, that the reputational penalty is unrelated to the size of the financial penalties levied (0.26% of the market capitalization on average). The results for France are all the more surprising that studies (on the US and other jurisdictions) concluded that financial and accounting issues-which are investigated by the article-triggered the strongest stock market reactions.⁵⁷ The reputational gain from being sanctioned echoes the initial statement that AMF sanctions seem misunderstood and neglected by analysts, investors, and shareholders. All in all, this article questions the information content of the sanctions, the credibility, usefulness, and efficiency of cash fines, and, more generally, of the current regulatory enforcement framework in France. Reputational penalties subsequent to sanctions could enhance enforcement, like in other jurisdictions. They could complement regulatory sanctions, if they were large enough to stand for a credible threat to offenders, without endangering the firms' solvency. The limited market reactions to AMF sanctions could also be due to the "person" being sanctioned: mostly companies, despite the frequent involvement of the top management

⁵⁷ In the US (Karpoff and Lott 1993; Griffin et al. 2004), but also in Japan and in China.

in the regulatory breaches (which would send a negative signal, according to the cross-sectional test results). Recent research suggests focusing more on top managers to gain in credibility and efficiency in deterring future crime (Jones 2013; Kay 2015; Cullen 2017). In the US, past research stressed how enforcement impacts carriers and reputations of top managers. Karpoff et al. (2008b) demonstrated that, in the US, 93% of top managers involved in financial misrepresentation lose their jobs before the end of the regulatory enforcement period, mostly explicitly fired. Complementarily, class-actions securities litigations will penalize directors' reputation only if initiated by the US SEC (Helland 2006). In the case of AMF sanctions, no top manager was fired, despite being involved in half of enforcement procedures of the sample.

Additional takeaways derive from the event-study analyses. The results confirmed past studies (Nourayi 1994; Alexander 1999; Murphy et al. 2009; Tibbs et al. 2011; Armour et al. 2017) in that financial wrongdoings linked to related parties induce stronger abnormal market reactions, in particular insider trading, and breaches to information obligations. It illustrates the key role played by trust in investment relationships. Sanctions seem to have gained in echo in the market since the Great Financial Crisis, implying higher abnormal returns, in line with Armour et al. (2017) in the UK, but contrary to Kirat and Rezaee (2019) for France.

Three remarks concern the transmission of the news of a regulatory sanction. Firstly, anonymizing the sanction report, when publishing it, appears to protect the sanctioned entity from suffering abnormal returns. Secondly, in line with past studies, a higher media coverage of the published sanction will trigger stronger abnormal negative returns. Thirdly, the cross-sectional results stress that the independence from governmental and political process (a key challenge for Regulators for Carvajal and Elliott (2007)) seems to be overcome. The successive chairmen of the AMF and of the AMF EC did not impact significantly market reactions.

In light of these results, some policy recommendations can be made in order to improve the credibility and the efficiency of the enforcement of financial regulation: (1) to increase transparency from part of the Regulator by communicating more on regulatory breaches, possibly before the sanction itself as done by the US SEC, as a tool to educate and set example ("name and shame"); (2) to sanction more severely (significantly higher fines, closer to the legal maximum), possibly less frequently, and to resort more disciplinary sanctions, if the Regulator believes that the credibility of a sanction should be measured in the market reactions; and (3) to focus more on individuals (top managers in particular), as a way to question their competences and integrity, possibly with higher fines or resorting to (temporary) bans on activity.

6 Conclusion

This work aimed at investigating the information content of sanctions of listed companies for financial misconducts, as enforced in France, to better understand how enforcement influences markets. It challenges the common view is that financial misconduct and regulatory breaches can be lightly punished by the French Regulator and consequently neglected by investors. It analyzes reactions in listed companies returns to the news of a sanction. The results are put into perspective depending on the content of the decision and on the characteristics of the convicts and compared with other jurisdictions. More precisely, this paper details the reactions of investors and stakeholders along the sanction proceeding by searching for abnormal returns after the four milestones of sanction procedures. It also strives to understand how the features of the sanctions, and of the sanctioned entities could explain such reactions. To do so, an original dataset was built for the 52 sanction decisions impacting 40 daily-listed companies from 2004 to 2016, completed with similar complementary datasets for acquittal decisions and settlements.

The results first show that the confidentiality of the early stages of the proceeding is respected, by the AMF and the investigated firms: no significant abnormal returns were detected. Investors then react significantly negatively to the news of a guilty sanction, and to its publication. Such negative abnormal returns are limited in absolute or relative terms, compared to past studies on France and on other jurisdictions. No reputational penalty is assorted to sanctions. Settlements do not trigger abnormal returns. Some features of the sanction and of the defendant will influence the reaction, but not the most straightforward (cash fine, behavioral sanction, and recidivism).

Overall, the results echo the reputation for leniency of AMF sanctions (scarce procedures, lax verdicts, low fines, ending neglected by investors), despite consecutive regulatory tightenings and long and costly procedures. They question the enforcement efficiency, bearing in mind that the goal of regulation is to support and accompany a healthy development of firms, and not to impose an unnecessary regulatory burden. The following policy recommendations can be made to improve the credibility of enforcement: (1) increasing the communication from the Regulator with more transparency on sanctions or with a name and shame of financial crime, for the market to quicker, better, and more comprehensively price the information; (2) sanctioning less frequently and more severely, with higher cash fines, and possibly completed with more frequent disciplinary sanctions⁵⁸; and (3) sanctioning more individuals, and in particular top managers.

Acknowledgements This work did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors. The Author wants to acknowledge the *Autorité des Marchés Financiers* (AMF) for giving access to regulatory information. Additionally, Catherine Bruneau, Gunther Capelle-Blancard, Cédric Euchenne, Evžen Kočenda, Fabrice Riva, and anonymous referees have to be thanked for their useful feedbacks and support, as well as the wide range of collaborators of the AMF, who accepted to be interviewed.

⁵⁸ The introduction of the settlements also supports that trend: dedicate more means to sanction the most severe regulatory breaches, by settling for the least serious—though significant—regulatory breaches. Indeed, since a peak in 2009, sanctions have been trending downwards. Sanctions hit a historical low point in 2016 and 2017 (15 per year, comparing with 23 on average per year over the 2004–2018 period), followed by a slight rebound in 2018 (20).

References

- Alexander, C. (1999). On the nature of the reputational penalty for corporate crime: evidence. *The Journal of Law and Economics*, 42(1), 489–526.
- Armour, J., Mayer, C., & Polo, A. (2017). Regulatory sanctions and reputational damage in financial markets. *Journal of Financial and Quantitative Analysis*, 52(4), 1429–1448.
- Barth, J. R., Caprio, G., & Levine, R. (2004). Bank supervision and regulation: what works best? Journal of Financial Intermediation, 13(2), 205–248.
- Becker, G. S. (1968). Crime and punishment: An economic approach. Journal of Political Economy, 76, 169–217.
- Beny, L. (2008). Do investors in controlled firms value insider trading laws? International evidence. Journal of Law, Economics and Policy, 4(2), 267–310.
- Berger, A., & Davies, S. (1998). The information content of bank examinations. Journal of Financial Services Research, 14(2), 117–144.
- Bryant, P. G., & Eckard, E. W. (1991). Price fixing: The probability of getting caught. *The Review of Economics and Statistics*, 73(3), 531–536.
- Bussmann, K. D., & Werle, M. M. (2006). Addressing crime in companies: First findings from a global survey of economic crime. *The British Journal of Criminology*, 46(6), 1128–1144.
- Caiazza, S., Cotugno, M., Fiordelisi, F., & Stefanelli, V. (2018). The spillover effect of enforcement actions on bank risk-taking. *Journal of Banking & Finance*, 91, 146–159.
- Carvajal, A., & Elliott, J. (2007). Strengths and weaknesses in securities market regulation: A global analysis. IMF Working Paper WP/07/259.
- Choi, S., & Kahan, M. (2007). Market penalty for mutual fund scandals. Boston University Law Review, 87, 1021–1057.
- Cohen, M. A. (1996). Theories of punishment and empirical trends in corporate criminal sanctions. *Managerial and Decision Economics*, 17(4), 399–411. Special Issue: Corporate Crime.
- Combe, E., Monier, C., & Legal, R. (2008). Cartels: The probability of getting caught in the european union. *Cahiers de recherche PRISM-Sorbonne*, no. 07-01-03.
- Cullen, J. (2017). A culture beyond repair? The Nexus between ethics and sanctions in finance. Just financial markets? Finance in a just society (pp. 154–184). Oxford: Oxford University Press.
- Cumming, D. J., & Johan, S. A. (2013). Listing standards and frauds. Managerial and Decision Economics, 34, 451–470.
- Cumming, D. J., Johan, S. A., & Li, D. (2011). Exchange trading rules and stock market liquidity. Journal of Financial Economics, 99(3), 651–671.
- de Batz, L. (2017a). Financial sanctions in France 2004–2016. Bankers, Markets & Investors, 149, 62-71.
- de Batz, L. (2017b). Lessons from more than a decade-long history of sanction in France? *LabEx Refi Working Paper Series*, no. 2017-11.
- de Batz, L. (2018). Financial impact of regulatory sanctions on French listed companies. *IES Working Paper*, no. 10/2018.
- Djama, C. (2008). Fraude à l'Information Comptable et Financière : le Rôle des Autorités de Régulation. Centre de Recherche en Management HAL-SHS, Working Paper n°00522510.
- Drake, M. S., Guest, N. M., & Twedt, B. J. (2014). The media and mispricing: The role of the business press in the pricing of accounting information. *The Accounting Review*, 89(5), 1673–1701.
- Dyck, A., Morse, A., & Zingales, L. (2010). Who blows the whistle on corporate fraud? *The Journal of Finance*, 65(6), 2213–2253.
- Engelen, P. J. (2011). What is the reputational cost of a dishonest CEO? Evidence from US illegal insider trading. CESifo Economic Studies, 58(1), 140–163.
- Fama, E. F., Fisher, L., Jensen, M. C., & Roll, R. (1969). The adjustment of stock prices to new information. *International Economic Review*, 10(1), 1–21.
- Fang, L. H., & Peress, J. (2009). Media coverage and the cross-section of stock returns. *The Journal of Finance*, 64(5), 2023–2052.
- Fang, L. H., Peress, J., & Zheng, L. (2014). Does media coverage of stocks affect mutual funds' trading and performance. *The Review of Financial Studies*, 27(12), 3441–3466.
- Feroz, E. H., Park, K., & Pastena, V. S. (1991). The financial and market effects of the SEC's accounting and auditing enforcement releases. *Journal of Accounting Research*, 29, 107–142.
- Fiordelisi, F., Soana, M. G., & Schwizer, P. (2014). Reputational losses and operational risk in banking. *The European Journal of Finance*, 20(2), 105–124.

- Fonteny, E. (2017). Les manquements d'Initiés en France. A Qui Profite le Crime? Revue Française de Gestion, 269, 39–63.
- Garoupa, N. (1999). Optimal law enforcement with dissemination of information. European Journal of Law and Economics, 7(3), 183–196.
- Gondhalekar, V., Joshi, M., & McKendall, M. (2012). Short-and long-term share price reaction to announcements of financial restatements. In *Advances in financial economics* (Vol. 15, pp. 149– 172). Emerald Group Publishing Limited.
- Griffin, P.A., Grundfest, J., & Perino, M.A. (2004). Stock price response to news of securities fraud litigation: An analysis of sequential and conditional information. *Abacus*, 40(1), 21–48.
- Haslem, B., Hutton, I., & Smith, A. H. (2017). How much do corporate defendants really lose? A new verdict on the reputation loss induced by corporate litigation. *Financial Management*, 46(2), 323–358.
- Heidinger, D., & Gatzert, N. (2018). Awareness, determinants and value of reputation risk management: Empirical evidence from the banking and insurance industry. *Journal of Banking & Finance*, 91, 106–118.
- Helland, E. (2006). Reputational penalties and the merits of class-action securities litigation. *The Journal* of Law and Economics, 49(2), 365–395.
- Jarrell, G., & Peltzman, S. (1985). The impact of product recalls on the wealth of sellers. Journal of Political Economy, 93(3), 512–536.
- Jones, R.M. (2013). Unfit for duty: The officer and director bar as a remedy for fraud. *Boston College Law School, Legal Studies Research paper Series*, Research paper no. 319.
- Karpoff, J.M., Koester, A., Lee, D.S., & Martin, G.S. (2014). Database challenges in financial misconduct research. *Georgetown University*, Working Paper no. 2012-15.
- Karpoff, J. M., Lee, D. S., & Martin, G. S. (2008a). The cost to firms of cooking the books. *Journal of Financial and Quantitative Analysis*, 43(3), 581–612.
- Karpoff, J. M., Lee, D. S., & Martin, G. S. (2008b). The consequences to managers for financial misrepresentation. *Journal of Financial Economics*, 88(88), 193–215.
- Karpoff, J. M., & Lott, J. R. (1993). Reputational penalty firms bear from committing criminal fraud. Journal of Law and Economics, 36(2), 757–802.
- Kay, J. (2015). Other people's money: Masters of the Universe of servants of the people?. London: Profile Books.
- Kirat, T., & Rezaee, A. (2019). How stock markets react to regulatory sanctions? Evidence from France. *Applied Economics*, 51(60), 6558–6566.
- Köster, H., & Pelster, M. (2017). Financial penalties and bank performance. Journal of Banking & Finance, 79(C), 57–73.
- Kothari, S. P., & Warner, J. B. (2008). Econometrics of event studies. In B. E. Eckbo (Ed.), Handbook of empirical corporate finance (pp. 3–36). Amsterdam: Elsevier.
- La Porta, R., Lopez de Silanes, F., & Shleifer, A. (2006). What works in securities laws? The Journal of Finance, 61(1), 1–32.
- Leng, F., Feroz, E., Cao, Z., & Davalos, S. (2011). The long-term performance and failure risk of firms cited in the US SEC's accounting and auditing enforcement releases. *Journal of Business Finance & Accounting*, 38(7–8), 813–841.
- Lin, J. C., & Rozeff, M. S. (1995). The speed of adjustment of prices to private information: Empirical tests. Journal of Financial Research, 18(2), 143–156.
- MacKinlay, A. C. (1997). Event studies in economics and finance. *Journal of Economic Literature*, 35(1), 13–39.
- Miller, G. S. (2006). The press as a watchdog for accounting fraud. *Journal of Accounting Research*, 44(5), 1001–1033.
- Murphy, D. L., Shrieves, R. E., & Tibbs, S. L. (2009). Understanding the penalties associated with corporate misconduct: An empirical examination of earnings and risk. *Journal of Financial and Quantitative Analysis*, 44(1), 55–83.
- Nelson, C., Gilley, S., & Trombley, G. (2009). Disclosures of SEC investigations resulting in wells notices. *Securities Litigation Journal*, 19(4), 19–21.
- Nourayi, M. M. (1994). Stock price responses to the SEC's enforcement actions. Journal of Accounting and Public Policy, 13(4), 333–347.
- Pritchard, A.C., & Ferris, S. P. (2001). Stock price reactions to securities fraud class actions under the private securities litigation reform act. *Michigan Law and Economics*, Research Paper no. 01-009.

- Rogers, J. L., Skinner, D. J., & Zechman, S. L. C. (2016). The role of the media in disseminating insidertrading news. *Review of Accounting Studies*, 21(3), 711–739.
- Tibbs, S. L., Harrell, D. H., & Shrieves, R. E. (2011). Do shareholders benefit from corporate misconduct? A long-run analysis. *Journal of Empirical Legal Studies*, 8(3), 449–476.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.